Author’s response to reviews

Title: Use of buccal mucosa grafts for urethral reconstruction in children: a retrospective cohort study

Authors:

Emilie K Johnson (emilie.johnson@childrens.harvard.edu)
Spencer I Kozinn (skozinn@yahoo.com)
Kathryn L Johnson (kathryn.johnson@childrens.harvard.edu)
Sohee Kim (skim113@gmail.com)
David A Diamond (david.diamond@childrens.harvard.edu)
Alan B Retik (alan.retik@childrens.harvard.edu)

Version: 2 Date: 28 February 2014

Author’s response to reviews:

Dear Editors, we have made edits detailed below. as requested prior to peer review of our manuscript. Thank you again for considering our paper for BMC Urology.

1. Name of the ethics committee approving our study was added to page 7, line 4-5.
2. There are no relevant acknowledgements. All people who contributed to this paper met criteria for authorship.
3. Page numbers were added to the manuscript.